Author’s response to reviews

Title: Determinants of Stunting among children aged 6 to 59 months in pastoral community, Afar Region, North East Ethiopia: Unmatched Case Control Study

Authors:
molla kahssay (mollaka2005@gmail.com)
Etsay Woldu (etsaywold@gmail.com)
Abel Gebre (abelgebre21@gmail.com)
P. Surender Reddy (dr.surenderreddy@yahoo.com)

Version: 4 Date: 01 Feb 2020

Author’s response to reviews:

Response to reviewer comments

Overall Response: Again, we authors are very grateful to your critical review you made so far. By now we have made a revision on the given comments and in overall contents of the manuscript. We critically see your previous comments and we have made a line by line revision of the manuscript. We authors are very thankful for your detailed and constructive review. We tried to revise the manuscript in terms of language and we did come up with the revised one. Hope language corrections we made are very essential. All comments are incorporated in the final version of this manuscript.

Editor Comments:
1. We note that the current submission contains some textual overlap with other previously published works. This overlap mainly exists in the Background section.

Response: Revision is made based on the given comments and incorporated in the final manuscript.

2. Please remove your "objective" subsection in your abstract. You may wish to combine the material with the "Background" subsection.

Response: Revision is made based on the given comments and incorporated in the final manuscript.

3. Please change the "introduction" heading to "background. And please change the "Methods and Materials" section to "methods"

Response: Revision is made based on the given comments and incorporated in the final manuscript.

4. We notice that in the Authors' Contribution section, author P. Surender Reddy is missing. Please include their name, in initial format, along with their individual contributions.

Response: Revision is made based on the given comments and incorporated in the final manuscript.

5. When submitting your revised manuscript please ensure you do so as a single clean copy without any tracked changes, colored or highlighted text, as these are no longer required at this stage of the editorial process.

Response: No highlighted and tracked changes in this final manuscript file.
THANK YOU!!!